Author’s response to reviews

Title: The use of CAM by women suffering from nausea and vomiting of pregnancy

Authors:

Taras Hollyer (thollyer@ccnm.edu)
Heather Boon (heather.boon@utoronto.ca)
Alexia Georgousis (ageorgousis@ccnm.edu)
Michael J Smith (michael_j_smith-@hc-sc.gc.ca)
Adrienne Einarson (einarson@sickkids.on.ca)

Version: 4 Date: 15 May 2002

My submission to BioMed central conforms to the checklist

signed
Adrienne Einarson